\documentclass[11pt]{article}

\usepackage{latexsym}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{mathrsfs}  
\usepackage{dsfont}    
\usepackage{bbold}     
\usepackage[english]{babel}
\usepackage{caption}
\usepackage{epsfig}
\usepackage{float}
%\usepackage{isolatin1}
\usepackage{subfigure}

\textwidth 16.cm
\textheight 22.2cm
\addtolength{\oddsidemargin}{-2.3cm}   
\addtolength{\evensidemargin}{-2.3cm}  
\addtolength{\topmargin}{-1.5cm}       

\sloppy

\setlength{\parskip}{0.5ex}             % Distancia entre paragrafos

\DeclareMathOperator{\Val}{Val}
\DeclareMathOperator{\med}{meas}

\newtheorem{theorem}{Theorem}
\newtheorem{theo}[theorem]{Theorem}
\newtheorem{coro}[theorem]{Corollary}
\newtheorem{defi}[theorem]{Definition}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{prop}[theorem]{Proposition}
\newtheorem{rmk}[theorem]{Remark}
\newtheorem{con}[theorem]{Conjecture}
\newtheorem{hyp}{Assumption}

% Teoremas contados por secao
\renewcommand{\thetheorem}{\thesection.\arabic{theorem}}

% Formulas contadas por secao
\renewcommand{\theequation}{\thesection.\arabic{equation}}

% Zera os contadores em nova secao
\newcommand{\zerarcounters}{\setcounter{equation}{0}\setcounter{theorem}{0}}

% Fontes pequenas nas legendas das figuras
\renewcommand{\captionfont}{\small}

% Conjunto de numeros
\newcommand{\ZZZ}{\mathds{Z}}
\newcommand{\CCC}{\mathds{C}}
\newcommand{\NNN}{\mathds{N}}
\newcommand{\QQQ}{\mathds{Q}}
\newcommand{\RRR}{\mathds{R}}
\newcommand{\TTT}{\mathds{T}}

% Caligraficos usuais
\newcommand{\calA}{{\mathcal A}}
\newcommand{\BB}{{\mathcal B}}
\newcommand{\CC}{{\mathcal C}}
\newcommand{\DD}{{\mathcal D}}
\newcommand{\calE}{{\mathcal E}}
\newcommand{\calF}{{\mathcal F}}
\newcommand{\calG}{{\mathcal G}}
\newcommand{\calH}{{\mathcal H}}
\newcommand{\calI}{{\mathcal I}}
\newcommand{\calJ}{{\mathcal J}}
\newcommand{\calK}{{\mathcal K}}
\newcommand{\calL}{{\mathcal L}}
\newcommand{\MM}{{\mathcal M}}
\newcommand{\calN}{{\mathcal N}}
\newcommand{\calO}{{\mathcal O}}
\newcommand{\calP}{{\mathcal P}}
\newcommand{\calQ}{{\mathcal Q}}
\newcommand{\RR}{{\mathcal R}}
\newcommand{\SSS}{{\mathcal S}}
\newcommand{\TT}{{\mathcal T}}
\newcommand{\calU}{{\mathcal U}}
\newcommand{\VV}{{\mathcal V}}
\newcommand{\calW}{{\mathcal W}}
\newcommand{\calX}{{\mathcal X}}
\newcommand{\calY}{{\mathcal Y}}
\newcommand{\calZ}{{\mathcal Z}}

% Caligraficos modificados 
\newcommand{\calmA}{{\mathscr A}}
\newcommand{\calmB}{{\mathscr B}}
\newcommand{\calmC}{{\mathscr C}}
\newcommand{\calmD}{{\mathscr D}}
\newcommand{\calmE}{{\mathscr E}}
\newcommand{\calmF}{{\mathscr F}}
\newcommand{\calmG}{{\mathscr G}}
\newcommand{\calmH}{{\mathscr H}}
\newcommand{\calmI}{{\mathscr I}}
\newcommand{\calmJ}{{\mathscr J}}
\newcommand{\calmK}{{\mathscr K}}
\newcommand{\calmL}{{\mathscr L}}
\newcommand{\calmM}{{\mathscr M}}
\newcommand{\calmN}{{\mathscr N}}
\newcommand{\calmO}{{\mathscr O}}
\newcommand{\calmP}{{\mathscr P}}
\newcommand{\calmQ}{{\mathscr Q}}
\newcommand{\calmR}{{\mathscr R}}
\newcommand{\calmS}{{\mathscr S}}
\newcommand{\calmT}{{\mathscr T}}
\newcommand{\calmU}{{\mathscr U}}
\newcommand{\calmV}{{\mathscr V}}
\newcommand{\calmW}{{\mathscr W}}
\newcommand{\calmX}{{\mathscr X}}
\newcommand{\calmY}{{\mathscr Y}}
\newcommand{\calmZ}{{\mathscr Z}}

% Goticos
\newcommand{\gota}{{\mathfrak a}}
\newcommand{\gotb}{{\mathfrak b}}
\newcommand{\gotc}{{\mathfrak c}}
\newcommand{\gotd}{{\mathfrak d}}
\newcommand{\gote}{{\mathfrak e}}
\newcommand{\gotf}{{\mathfrak f}}
\newcommand{\gotg}{{\mathfrak g}}
\newcommand{\goth}{{\mathfrak h}}
\newcommand{\goti}{{\mathfrak i}}
\newcommand{\gotj}{{\mathfrak j}}
\newcommand{\gotk}{{\mathfrak k}}
\newcommand{\gotl}{{\mathfrak l}}
\newcommand{\gotm}{{\mathfrak m}}
\newcommand{\gotn}{{\mathfrak n}}
\newcommand{\goto}{{\mathfrak o}}
\newcommand{\gotp}{{\mathfrak p}}
\newcommand{\gotq}{{\mathfrak q}}
\newcommand{\gotr}{{\mathfrak r}}
\newcommand{\gots}{{\mathfrak s}}
\newcommand{\gott}{{\mathfrak t}}
\newcommand{\gotu}{{\mathfrak u}}
\newcommand{\gotv}{{\mathfrak v}}
\newcommand{\gotx}{{\mathfrak x}}
\newcommand{\gotz}{{\mathfrak z}}
\newcommand{\gotw}{{\mathfrak w}}
\newcommand{\goty}{{\mathfrak y}}
\newcommand{\gotA}{{\mathfrak A}}
\newcommand{\gotB}{{\mathfrak B}}
\newcommand{\gotC}{{\mathfrak C}}
\newcommand{\gotD}{{\mathfrak D}}
\newcommand{\gotE}{{\mathfrak E}}
\newcommand{\gotF}{{\mathfrak F}}
\newcommand{\gotG}{{\mathfrak G}}
\newcommand{\gotH}{{\mathfrak H}}
\newcommand{\gotI}{{\mathfrak I}}
\newcommand{\gotJ}{{\mathfrak J}}
\newcommand{\gotK}{{\mathfrak K}}
\newcommand{\gotL}{{\mathfrak L}}
\newcommand{\gotM}{{\mathfrak M}}
\newcommand{\gotN}{{\mathfrak N}}
\newcommand{\gotO}{{\mathfrak O}}
\newcommand{\gotP}{{\mathfrak P}}
\newcommand{\gotQ}{{\mathfrak Q}}
\newcommand{\gotR}{{\mathfrak R}}
\newcommand{\gotS}{{\mathfrak S}}
\newcommand{\gotT}{{\mathfrak T}}
\newcommand{\gotU}{{\mathfrak U}}
\newcommand{\gotV}{{\mathfrak V}}
\newcommand{\gotX}{{\mathfrak X}}
\newcommand{\gotZ}{{\mathfrak Z}}
\newcommand{\gotW}{{\mathfrak W}}
\newcommand{\gotY}{{\mathfrak Y}}

% Matematicos
\newcommand{\matA}{{\mathscr A}}
\newcommand{\matB}{{\mathscr B}}
\newcommand{\matC}{{\mathscr C}}
\newcommand{\matD}{{\mathscr D}}
\newcommand{\matE}{{\mathscr E}}
\newcommand{\matF}{{\mathscr F}}
\newcommand{\matG}{{\mathscr G}}
\newcommand{\matH}{{\mathscr H}}
\newcommand{\matI}{{\mathscr I}}
\newcommand{\matJ}{{\mathscr J}}
\newcommand{\matK}{{\mathscr K}}
\newcommand{\matL}{{\mathscr L}}
\newcommand{\matM}{{\mathscr M}}
\newcommand{\matN}{{\mathscr N}}
\newcommand{\matO}{{\mathscr O}}
\newcommand{\matP}{{\mathscr P}}
\newcommand{\matQ}{{\mathscr Q}}
\newcommand{\matR}{{\mathscr R}}
\newcommand{\matS}{{\mathscr S}}
\newcommand{\matT}{{\mathscr T}}
\newcommand{\matU}{{\mathscr U}}
\newcommand{\matV}{{\mathscr V}}
\newcommand{\matX}{{\mathscr X}}
\newcommand{\matZ}{{\mathscr Z}}
\newcommand{\matW}{{\mathscr W}}
\newcommand{\matY}{{\mathscr Y}}

% Comandos para sublinhar letras (em math mode)
\newcommand{\und}{\underline}
\newcommand{\unda}{\underline{a}}
\newcommand{\undb}{\underline{b}}
\newcommand{\undc}{\underline{c}}
\newcommand{\undd}{\underline{d}}
\newcommand{\unde}{\underline{e}}
\newcommand{\undf}{\underline{f}}
\newcommand{\undg}{\underline{g}}
\newcommand{\undh}{\underline{h}}
\newcommand{\undi}{\underline{i}}
\newcommand{\undj}{\underline{j}}
\newcommand{\undk}{\underline{k}}
\newcommand{\undl}{\underline{l}}
\newcommand{\undm}{\underline{m}}
\newcommand{\undn}{\underline{n}}
\newcommand{\undo}{\underline{o}}
\newcommand{\undp}{\underline{p}}
\newcommand{\undq}{\underline{q}}
\newcommand{\undr}{\underline{r}}
\newcommand{\unds}{\underline{s}}
\newcommand{\undt}{\underline{t}}
\newcommand{\undu}{\underline{u}}
\newcommand{\undv}{\underline{v}}
\newcommand{\undw}{\underline{w}}
\newcommand{\undx}{\underline{x}}
\newcommand{\undy}{\underline{y}}
\newcommand{\undz}{\underline{z}}
\newcommand{\undA}{\underline{A}}
\newcommand{\undB}{\underline{B}}
\newcommand{\undC}{\underline{C}}
\newcommand{\undD}{\underline{D}}
\newcommand{\undE}{\underline{E}}
\newcommand{\undF}{\underline{F}}
\newcommand{\undG}{\underline{G}}
\newcommand{\undH}{\underline{H}}
\newcommand{\undI}{\underline{I}}
\newcommand{\undJ}{\underline{J}}
\newcommand{\undK}{\underline{K}}
\newcommand{\undL}{\underline{L}}
\newcommand{\undM}{\underline{M}}
\newcommand{\undN}{\underline{N}}
\newcommand{\undO}{\underline{O}}
\newcommand{\undP}{\underline{P}}
\newcommand{\undQ}{\underline{Q}}
\newcommand{\undR}{\underline{R}}
\newcommand{\undS}{\underline{S}}
\newcommand{\undT}{\underline{T}}
\newcommand{\undU}{\underline{U}}
\newcommand{\undV}{\underline{V}}
\newcommand{\undW}{\underline{W}}
\newcommand{\undX}{\underline{X}}
\newcommand{\undY}{\underline{Y}}
\newcommand{\undZ}{\underline{Z}}
\newcommand{\undzero}{\underline{0}}
\newcommand{\undomega}{\underline{\omega}}

% Shortcuts
\newcommand{\Fullbox}{{\rule{2.0mm}{2.0mm}}}
\newcommand{\ES}{\hfill\Box}
\newcommand{\EP}{\hfill\Fullbox\vspace{0.2cm}}
\newcommand{\prova}{\noindent{\it Proof. }}
\newcommand{\io}{\infty}
\newcommand{\e}{\varepsilon}
\newcommand{\al}{\alpha}
\newcommand{\be}{\beta}
\newcommand{\n}{\nu}
\newcommand{\m}{\mu}
\newcommand{\x}{\xi}
\newcommand{\p}{\pi}
\newcommand{\ka}{\kappa}
\newcommand{\g}{\gamma}
\newcommand{\om}{\omega}
\newcommand{\la}{\lambda}
\newcommand{\f}{\varphi}
\newcommand{\s}{\sigma}
\newcommand{\UM}{\mathbb{1}}
\newcommand{\del}{\partial}
\newcommand{\laa}{\langle}
\newcommand{\raa}{\rangle}
\newcommand{\Me}[1]{\laa#1\raa}
\newcommand{\pint}[2]{\langle#1, \, #2\rangle}
\newcommand{\av}[1]{\langle #1 \rangle}
\newcommand{\wh}{\widehat}
\newcommand{\vm}{\mathbf{m}}
\newcommand{\oo}{\boldsymbol{\omega}}
\newcommand{\nn}{\boldsymbol{\nu}}
\newcommand{\pps}{\boldsymbol{\psi}}
\newcommand{\vzero}{\boldsymbol{0}}
\newcommand{\der}{{\rm d}}
\newcommand{\ii}{{\rm i}}

\def\ins#1#2#3{\vbox to0pt{\kern-#2 \hbox{\kern#1 #3}\vss}\nointerlineskip}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}

{\small \bibitem{BDG}{
M.V. Bartuccelli, J.H.B. Deane, G. Gentile,
\textit{Globally and locally attractive solutions
for quasi-periodically forced systems},
J. Math. Anal. Appl.
\textbf{328} (2007), no. 1, 699-714. }
%
\bibitem{BDGM}{
M.V. Bartuccelli, J.H.B. Deane, G. Gentile, L. Marsh,
\textit{Invariant sets for the varactor equation},
Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci.
\textbf{462} (2006), no. 2066, 439-457. }
%
\bibitem{BHS}{
H.W. Broer, G.B. Huitema, M.B. Sevryuk,
\textit{Quasi-periodic motions in families of dynamical systems.
Order amidst chaos},
Lecture Notes in Mathematics 1645,
Springer, Berlin, 1996. }
%
\bibitem{B}{
A.D. Bryuno,
\textit{Analytic form of differential equations. I, II},
Trudy Moskov. Mat. Ob\v{s}\v{c}.
\textbf{25} (1971), 119--262;
ibid. \textbf{26} (1972), 199--239. 
English translations:
\textit{Trans. Moscow Math. Soc.}
\textbf{25} (1971), 131--288 (1973);
ibid. \textbf{26} (1972), 199--239 (1974). } 
%
\bibitem{CG}{
L. Corsi, G. Gentile,
\textit{Melnikov theory to all orders and Puiseux series
for subharmonic solutions},
J. Math. Phys.
\textbf{49} (2008), no. 11, 112701, 29 pp. }
%
\bibitem{CGGG}{
O. Costin, G. Gallavotti, G. Gentile, A. Giuliani,
\textit{Borel summability and Lindstedt series},
Comm. Math. Phys.
\textbf{269} (2007), no. 1, 175-193. }
%
\bibitem{GBD1}{
G. Gentile, M.V. Bartuccelli, J.H.B. Deane,
\textit{Summation of divergent series and Borel summability
for strongly dissipative differential equations with periodic
or quasiperiodic forcing terms},
J. Math. Phys. \textbf{46} (2005), no. 6, 062704, 21 pp. }
%
\bibitem{GBD2}{
G. Gentile, M.V. Bartuccelli, J.H.B. Deane,
\textit{Quasiperiodic attractors, Borel summability
and the Bryuno condition for strongly dissipative systems},
J. Math. Phys. \textbf{47} (2006), no. 7, 072702, 10 pp. }
%
\bibitem{G1}{
G. Gentile,
\textit{Quasi-periodic solutions for two-level systems},
Comm. Math. Phys.
\textbf{242} (2003),  no. 1-2, 221--250. }
%
\bibitem{G2}{
G. Gentile,
\textit{Resummation of perturbation series and reducibility
for Bryuno skew-product flows},
J. Stat. Phys.
\textbf{125} (2006), no. 2, 321-361. }
%
\bibitem{G3}{
G. Gentile,
\textit{Degenerate lower-dimensional tori under the Bryuno condition},
Ergodic Theory and Dynamical Systems,
\textbf{27} (2007), no. 2, 427-457. }
%
\bibitem{L}{
J. Lopes Dias,
\textit{A normal form theorem for Brjuno skew systems through renormalization},
J. Differential Equations
\textbf{230} (2006), no. 1, 1--23. }
%
\bibitem{MMY}{
S. Marmi, P. Moussa, J.-Ch. Yoccoz,
\textit{Some properties of real and complex Brjuno functions},
Frontiers in number theory, physics, and geometry. I,
pp. 601--623, Springer, Berlin, 2006. }
%
\bibitem{S}{
A.D. Sokal,
\textit{An improvement of Watson's theorem on Borel summability},
J. Math. Phys. \textbf{21} (1980), no. 2, 261--263. }
%
\bibitem{P}{
M. Ponce,
\textit{On the persistence of invariant curves for fibered
holomorphic transformations}, 
Preprint, 2008 (http://arxiv.org/abs/0801.2549). }
%
\bibitem{St}{
J.J. Stoker,
\textit{Nonlinear Vibrations in Mechanical and Electrical Systems},
Interscience Publishers, Inc., New York, N.Y., 1950. }

}

\end{thebibliography}


\title{\bf Quasi-periodic motions\\in
strongly dissipative forced systems}

\author
{\bf Guido Gentile
\vspace{2mm}
\\ \small 
$^\dagger$Dipartimento di Matematica, Universit\`a di Roma Tre, Roma,
I-00146, Italy.
\\ \small 
E-mail: gentile@mat.uniroma3.it}

\date{}

\maketitle

\begin{abstract}
We consider a class of ordinary differential equations describing
one-dimensional systems with a quasi-periodic forcing term
and in the presence of large damping. We discuss the conditions
to be assumed on the mechanical force and the forcing term
for the existence of quasi-periodic solutions which have
the same frequency vector as the forcing.
\end{abstract}

%\newpage

%\tableofcontents

%\newpage

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{Introduction}
\label{sec:1}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this paper we study the same problem considered in \cite{GBD1,GBD2},
that is the existence of quasi-periodic motions in strongly dissipative
forced systems, with the aim of removing as far as possible the
non-degeneracy condition on the mechanical force and the forcing.

We consider one-dimensional systems with a quasi-periodic forcing
term in the presence of strong damping, described
by ordinary differential equations of the form
%
\begin{equation}
\e \ddot x + \dot x + \e g(x) = \e f(\oo t) ,
\label{eq:1.1} \end{equation}
%
where $g(x)$ is the \textit{mechanical force},
$f(\oo t)$ is the \textit{forcing term},
$\oo\in\RRR^{d}$ is the \textit{frequency vector} of the forcing,
and $\g=1/\e>0$ is the \textit{damping coefficient}.
Systems of the form (\ref{eq:1.1}) naturally arise in
classical mechanics and electronic engineering;
we refer to \cite{BDGM,GBD1} for physical motivations.
A classical question in the case of forced systems asks for
\textit{response solutions}, that is solutions which are quasi-periodic
with the same frequency vector as the forcing. Note that in
(\ref{eq:1.1}) the forcing is not assumed to be small, as usually
done \cite{St} (see also \cite{BHS} for a review of recent developments):
it is the inverse of the damping coefficient which plays
the role of the perturbation parameter.

Both functions $g$ and $f$ will be assumed to be analytic
in their arguments, with $f$ quasi-periodic, i.e.
%
\begin{equation}
f(\pps) = \sum_{\nn\in \ZZZ^{d}}
{\rm e}^{\ii\nn \cdot \pps} f_{\nn} , \qquad \pps \in \TTT^{d} ,
\label{eq:1.2} \end{equation}
%
with average $\laa f \raa = f_{\vzero}$, and
$\cdot$ denoting the scalar product in $\RRR^{d}$.
By the analyticity assumption on $f$, one has
$|f_{\nn}| \le \Phi {\rm e}^{-\xi|\nn|}$
for suitable positive constants $\Phi$ and $\xi$.

A Diophantine condition is assumed on $\oo$. Define
the \textit{Bryuno function} \cite{B}
%
\begin{equation}
\gotB(\oo) = \sum_{n=0}^{\infty} \frac{1}{2^{n}} \log
\frac{1}{\al_{n}(\oo)} , \qquad \al_{n}(\oo) = \inf
\{ |\oo\cdot\nn| : \nn \in \ZZZ^{d}
\hbox{ such that } 0<|\nn|\le 2^{n} \} .
\label{eq:1.3} \end{equation}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{hyp}\label{hyp:1} 
The frequency vector $\oo$ satisfies
the Bryuno condition $\gotB(\oo)<\infty$.
\end{hyp}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Note that if $\oo\in\RRR^{d}$ satisfies the
standard Diophantine condition
$\left|\oo\cdot\nn \right| \ge \g_{0} |\nn|^{-\tau}$
for all $\nn\in\ZZZ^{d}_{*}$,
where $|\nn|:=|\nn|_{1}\equiv|\n_{1}|+\ldots+|\n_{d}|$ and
$\ZZZ^{d}_{*} := \ZZZ^{d} \setminus\{\vzero\}$, and
for some positive constants $\g_{0}$ and $\tau$, then
it also satisfies (\ref{eq:1.3}).
Recently, the Bryuno condition has received a lot of attention
in the theory of small divisor problems; see for instance
\cite{MMY,G2,G3,L,P} and papers cited therein.

The following assumption will be made on the functions $g$ and $f$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{hyp}\label{hyp:2} 
There exists $c_{0}\in\RRR$ such that $x=c_{0}$ is a zero
of odd order $\gotn$ of the equation
%
\begin{equation}
g(x) - f_{\vzero} = 0 ,
\label{eq:1.4} \end{equation}
%
that is $\der^{\gotn}g/\der x^{\gotn}(c_{0})\neq0$ and, if $\gotn>1$,
$\der^{k} g/\der x^{k}(c_{0})=0$ for $k=1,\ldots,\gotn -1$.
\end{hyp}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Of course, for given force $g(x)$, one can read Assumption \ref{hyp:2}
as a condition on the forcing term.

In \cite{GBD1,GBD2} we considered Assumption \ref{hyp:2}
with $\gotn=1$, and, in that case, we proved that
for $\e>0$ small enough there exists a quasi-periodic solution
with frequency vector $\oo$, reducing to $c_{0}$ as $\e$ tends to $0$,
and that such a solution is analytic in a circle tangent
at the origin to the vertical axis.

In this paper we show that the same result of existence extends
under the weaker Assumption \ref{hyp:2}. We also show
that in the case of even $\gotn$ a quasi-periodic solution
oscillating around $x=c_{0}$ fails to exist.
More formal statements are given in Section \ref{sec:2}.

The paper is organised as follows. In Section \ref{sec:2}
we split the problem into two equations, which, using
standard terminology, will be called the \textit{range equation}
and the \textit{bifurcation equation}. The first one involves
small denominator problems, and will be solved iteratively
in Section \ref{sec:3} by using techniques of multiscale analysis
\cite{G1,G2,G3}; from a technical point of view this is the core
of the paper. The second one is an implicit function equation,
and will be discussed in Section \ref{sec:4}.
In Section \ref{sec:5} we show that in the case of zeroes
of even order for the equation (\ref{eq:1.4}), a quasi-periodic
solution of the form $x(t)=c_{0}+O(\e)$ does not exist.
Finally in Section \ref{sec:6} we draw some conclusions and remarks.
The paper is fully self-contained,
and no acquaintance with previous works is required.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{Setting the problem}
\label{sec:2}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We are interested in the existence of a quasi-periodic solution
with frequency vector $\oo$, hence we expand $x$ as
%
\begin{equation}
x(t) = c + X(\oo t) , \qquad
X(\pps) = \sum_{\nn\in\ZZZ^{d}_{*}} {\rm e}^{\ii\nn\cdot\pps} X_{\nu} ,
\label{eq:2.1} \end{equation}
%
where $c=x_{\vzero}$ is the average of $x$ (hence $X$ is
a zero-average function).
Thus, we can rewrite (\ref{eq:1.1}) in Fourier space as
%
\begin{equation}
\begin{cases}
\left( \ii\oo\cdot\nn \right)
\left( 1 + \ii\e\oo\cdot\nn \right) X_{\nn} +
\e [g \circ (c + X)]_{\nn} = \e f_{\nn} , & \nn \neq \vzero , \\
[g \circ (c + X)]_{\vzero} = f_{\vzero} , & \nn =\vzero ,
\end{cases}
\label{eq:2.2} \end{equation}
%
where $[F]_{\nn}$ denotes the $\nn$-th Fourier coefficient
of the function $F$.

We shall adopt the following strategy. We shall look for
a solution $c+X$ of the range equation, i.e. the first equation
in (\ref{eq:2.2}), with $c$ arbitrary, and thereafter we shall fix $c$
in such a way that the bifurcation equation, i.e. the second equation
in (\ref{eq:2.2}), be satisfied. This suggests us to consider,
besides the equations (\ref{eq:2.2}), also the equation
%
\begin{equation}
\left( \ii\oo\cdot\nn \right)
\left( 1 + \ii\e\oo\cdot\nn \right) X_{\nn} +
\e [g \circ (c + X)]_{\nn} = \e f_{\nn} , \qquad \nn \neq \vzero ,
\label{eq:2.3} \end{equation}
%
and we shall look for a solution of the form (\ref{eq:2.1}) to
(\ref{eq:2.3}). In Section \ref{sec:3} we shall prove that, for
any $c\in\RRR$ close enough to $c_{0}$ and all $\e$ small enough,
there exist such a solution $x=c+X(\oo t;\e,c)$.
Then in Section \ref{sec:4} we shall study the bifurcation equation
%
\begin{equation}
[g(c+X(\cdot;\e,c)]_{\vzero} = f_{\vzero} ,
\label{eq:2.4} \end{equation}
%
and we shall see that for $\e$ small enough there exists a solution $c$
to (\ref{eq:2.4}), tending to $c_{0}$ as $\e$ tends to $0$.
More precisely we shall prove the following result.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{theorem} \label{thm:1}
Under the Assumptions \ref{hyp:1} and \ref{hyp:2} for the
ordinary differential equation (\ref{eq:1.1}), for all
$\e$ small enough there exist a continuous function $c(\e)$ and
a response solution $x(t)=c(\e)+X(\oo t;\e,c(\e))$ to (\ref{eq:1.1}),
with $c(0)=c_{0}$ and the function $X(\pps;\eta,c)$ which is
$C^{\io}$ in $\eta$ and $c$, vanishing at $\eta=0$,
and $2\pi$-periodic, analytic and zero-average in $\pps$.
\end{theorem}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

If $c_{0}$ is not a zero of the equation (\ref{eq:1.4}), obviously
there is no quasi-periodic solution to (\ref{eq:1.1}) reducing to
$c_{0}$ as $\e$ tends to $0$. We shall show that the same non-existence
result holds if $c_{0}$ is a zero of even order of (\ref{eq:1.4}).
Therefore, the following result strengthens Theorem \ref{thm:1}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{theorem} \label{thm:2}
Under Assumption \ref{hyp:1} for the ordinary differential equation
(\ref{eq:1.1}), there exists a quasi-periodic solution
of the form as in Theorem \ref{thm:1} if and only if
Assumption \ref{hyp:2} is satisfied. In particular if $c_{0}$ is a
zero of even order of (\ref{eq:1.4}) such a solution does not exists.
\end{theorem}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{The small denominator equation and multiscale analysis}
\label{sec:3}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

A graph is a connected set of points and lines.
A \textit{tree} $\theta$ is a graph with no cycle,
such that all the lines are oriented toward a unique
point (\textit{root}) which has only one incident line (root line).
All the points in a tree except the root are called \textit{nodes}.
The orientation of the lines in a tree induces a partial ordering 
relation ($\preceq$) between the nodes. Given two nodes $v$ and $w$,
we shall write $w \prec v$ every time $v$ is along the path
(of lines) which connects $w$ to the root.

We call $E(\theta)$ the set of \textit{end nodes} in $\theta$,
that is the nodes which have no entering line, and $V(\theta)$
the set of \textit{internal nodes} in $\theta$, that is the set of
nodes which have at least one entering line. Set $N(\theta)=
E(\theta) \amalg V(\theta)$. With each end node $v$ we associate
a \textit{mode} label $\nn_{v}\in\ZZZ^{d}_{*}$. For all $v\in N(\theta)$
denote with $s_{v}$ the number of lines entering the node $v$.

We denote with $L(\theta)$ the set of lines in $\theta$. Since a
line $\ell$ is uniquely identified with the node $v$ which it leaves,
we may write $\ell = \ell_{v}$. With each line $\ell$ we associate
a \textit{momentum} label $\nn_{\ell} \in \ZZZ^{d}_{*}$ and
a \textit{scale} label $n_{\ell}\in\ZZZ_{+}$.

The modes of the end nodes and and the momenta of the lines
are related as follows: if $\ell = \ell_{v}$ one has
%
\begin{equation}
\nn_{\ell} = \sum_{w \in E(\theta) : w \preceq v} \nn_{w} .
\label{eq:3.1} \end{equation}
%
If $v$ is an internal node then (\ref{eq:3.1}) gives
$\nn_{\ell}=\nn_{\ell_{1}}+\ldots+\nn_{\ell_{s_{v}}}$, where
$\ell_{1},\ldots,\ell_{s_{v}}$ are the lines entering $v$.

We call \textit{equivalent} two trees which can be transformed into
each other by continuously deforming the lines in such a way that
they do not cross each other. Let $\TT_{k,\nn}$ be the set of
inequivalent trees of order $k$ and total momentum $\nn$,
that is the set of inequivalent trees $\theta$ such that $|N(\theta)|=
|V(\theta)|+|E(\theta)|=k$ and the momentum of the root line is $\nn$.

A cluster $T$ on scale $n$ is a maximal set of nodes and lines
connecting them such that all the lines have scales $n'\le n$
and there is at least one line with scale $n$. The lines entering
the cluster $T$ and the possible line coming out from it (unique if
existing at all) are called the external lines of the cluster $T$.
Given a cluster $T$ on scale $n$, we shall denote by $n_{T}=n$ the
scale of the cluster. We call $V(T)$, $E(T)$, and $L(T)$ the set of
internal nodes, of end nodes, and of lines of $T$, respectively;
note that the external lines of $T$ do not belong to $L(T)$.

We call self-energy cluster any cluster $T$ such that
$T$ has only one entering line $\ell_{T}^{2}$ and one exiting
line $\ell_{T}^{1}$, and one has $\sum_{v\in E(T)} \nn_{v} = \vzero$
(and hence $\nn_{\ell_{T}^{1}}=\nn_{\ell_{T}^{2}}$).
Call $\calP_{T}$ the path of lines $\ell\in L(T)$ connecting
$\ell_{T}^{2}$ to $\ell_{T}^{1}$, and set $x_{T}=
\oo\cdot\nn_{\ell_{T}^{1}}=\oo\cdot\nn_{\ell_{T}^{2}}$.
Let $\gotT_{k,\nn}$ be the set of \textit{renormalised trees} in
$\TT_{k,\nn}$, i.e. of trees in $\TT_{k,\nn}$ which do not
contain any self-energy clusters.

If we write
%
\begin{equation}
g(x) = \sum_{s=0}^{\io} g_{s}(c) (x-c)^{s} ,
\qquad g_{s}(c) := \frac{1}{s!} \frac{\der^{s}}{\der x^{s}} g(c) ,
\label{eq:3.2} \end{equation}
%
then we can choose $r>0$ such that $|g_{s}(c)| \le \Gamma^{s}$ for
all $c\in B_{r}(c_{0})$, with the constant $\Gamma$ independent of $c$.

Let $\psi$ be a non-decreasing $C^{\infty}$ function defined
in $\RRR_{+}$, such that
%
\begin{equation}
\psi(u) = \left\{
\begin{array}{ll}
1 , & \text{for } u \geq 1 , \\
0 , & \text{for } u \leq 1/2 ,
\end{array} \right.
\label{eq:3.3} \end{equation}
%
and set $\chi(u) := 1-\psi(u)$. For all $n \in \ZZZ_{+}$ define
$\chi_{n}(u) := \chi(u/4\al_{n}(\oo))$ and $\psi_{n}(u) :=
\psi(u/4\al_{n}(\oo))$, and set
%
\begin{equation}
\Xi_{n}(x)=\chi_{0}(|x|)\ldots \chi_{n-1}(|x|) \chi_{n}(|x|) , \qquad
\Psi_{n}(x)=\chi_{0}(|x|)\ldots \chi_{n-1}(|x|) \psi_{n}(|x|) .
\label{eq:3.4} \end{equation}
%

We associate with each node $v$ a \textit{node factor}
%
\begin{equation}
F_{v} = \begin{cases}
- \displaystyle{ \frac{1}{s_{v}!} g_{s_{v}}(c) } ,
& v \in V(\theta) , \\
f_{\nn_{v}} , & v \in E(\theta) , \end{cases}
\label{eq:3.5} \end{equation}
%
and we associate with each line $\ell$ a \textit{propagator}
%
\begin{equation}
G_{\ell} = G^{[n_{\ell}]}(\oo\cdot\nn_{\ell};\e,c) ,
\label{eq:3.6} \end{equation}
%
where the functions $G^{[n]}(x;\e,c)$
are recursively defined for $n\ge 0$ as
%
\begin{subequations}
\begin{align}
& \hskip-.3truecm
G^{[n]}(x;\e,c) =
\frac{\Psi_{n}(x)}{ \ii x(1+\ii \e x) - \MM^{[n-1]}(x;\e,c)} ,
\label{eq:3.7a} \\
& \hskip-.3truecm
\MM^{[n]}(x;\e,c) = \MM^{[n-1]}(x;\e,c) +
\Xi_{n}(x)
M^{[n]}(x;\e,c) , \quad
M^{[n]}(x;\e,c) = \!\! \sum_{T\in\gotR_{n}}\Val(T,x;\e,c) ,
\label{eq:3.7b}
\end{align}
\label{eq:3.7} \end{subequations}
%
\vskip-.3truecm
\noindent where $\MM^{[-1]}(x;\e,c)=\e g_{1}(c)$, $\gotR_{n}$ is the set
of renormalised self-energy clusters, i.e. of self-energy clusters which
do not contain any further self-energy clusters, on scale $n$, and
%
\begin{equation}
\Val(T,x;\e,c) = \Big(\prod_{\ell \in L(T)} G_{\ell} \Big) 
\Big( \prod_{v \in N(T)} F_{v} \Big)
\label{eq:3.8} \end{equation}
%
is called the value of the self-energy cluster $T$.
Note that $\MM^{[-1]}(x;\e,c)=0$ for $\gotn>1$ in Assumption \ref{hyp:2}.

Set
%
\begin{equation}
X_{\nn}^{[k]} = \sum_{\theta \in \gotT_{k,\n}}
\Val(\theta;\e,c) , \qquad
\Val(\theta;\e,c) = \Big(\prod_{\ell \in L(\theta)} G_{\ell} \Big) 
\Big( \prod_{v \in N(\theta)} F_{v} \Big) ,
\label{eq:3.9} \end{equation}
%
where $\Val(\theta;\e,c)$ is called the value of the tree $\theta$,
and define the \textit{renormalised  series}
%
\begin{equation}
\overline X(\pps;\e,c)= \sum_{\nn\in\ZZZ^{d}_{*}}
{\rm e}^{\ii\nn\cdot\pps} \overline X_{\nn} ,
\qquad \overline X_{\nn} = \sum_{k=1}^{\io} \e^{k} X_{\nn}^{[k]} .
\label{eq:3.10} \end{equation}
%
Set also
%
\begin{equation}
M(\theta) = \sum_{v \in E(\theta)} |\nn_{v}| , \qquad
M(T) = \sum_{v \in E(T)} |\nn_{v}| , \qquad
\label{eq:3.11} \end{equation}
%
and call $\gotN_{n}(\theta)$ the number of lines $\ell\in L(\theta)$
such that $n_{\ell}\ge n$, and $\gotN_{n}(T)$ the number of lines
$\ell\in L(T)$ such that $n_{\ell}\ge n$.

Finally define
%
\begin{equation}
n(\nn) = \inf \left\{ n \in \ZZZ_{+} : |\nn| \le 2^{n} \right\} .
\label{eq:3.12} \end{equation}
%
Note that $|\oo\cdot\nn| \ge \al_{n(\nn)}(\oo)$, and 
$\al_{n'}(\oo)<\al_{n}(\oo)$ implies $n'>n$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:1}
For any renormalised tree $\theta$,
one has $\gotN_{n}(\theta) \le 2^{-(n-2)}M(\theta)$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova We prove that $\gotN_{n}(\theta) \le \max \{ 0, 2^{-(n-2)}
M(\theta)-1\}$ by induction on the number of nodes of $\theta$.
If $N(\theta)=1$ and $\gotN_{n}(\theta)=1$, then $\theta$
has only one line $\ell$ and $n_{\ell}\ge n$. Thus,
$|\oo\cdot\nn_{\ell}| \le \al_{n-1}(\oo)/4$, so that
$n(\nn_{\ell}) \ge n$, and hence $|\nn_{\ell}| > 2^{n-1}$,
which implies $2^{-(n-2)}M(\theta)=2^{-(n-2)}|\nn_{\ell}| \ge 2$.

If $N(\theta) > 1$, let $\ell_{0}$ be the root line of $\theta$
and set $\nn=\nn_{\ell_{0}}$. If $n_{\ell_{0}} < n$ the assertion follows
from the inductive hypothesis.
If $n_{\ell_{0}}\ge n$, call $\ell_{1},\ldots,\ell_{m}$ the lines
with scale $\ge n$ which are closest to $\ell_{0}$. The case $m=0$
is trivial. If $m\ge 2$ the bound follows once more from the
inductive hypothesis. Finally, if $m=1$, then $\ell_{1}$ is
the entering line of a cluster $T$ and $\nn'\neq \nn$,
where $\nn'= \nn_{\ell_{1}}$.
Then $|\oo\cdot(\nn-\nn')| \le \al_{n-1}(\oo)/2$, so that
$n(\nn-\nn')\ge n-1$, and hence $M(T)\ge |\nn-\nn'| > 2^{n-2}$.
Therefore, if $\theta_{1}$ is the tree with root line $\ell_{1}$,
one has $M(\theta)=M(T)+M(\theta_{1})$ and hence
%
\begin{equation}
\gotN_{n}(\theta) = 1 + \gotN_{n}(\theta_{1}) \le
2^{-(n-2)}M(\theta_{1}) \le
2^{-(n-2)}M(\theta) - 2^{-(n-2)}M(T) \le 2^{-(n-2)}M(\theta) - 1 .
\nonumber \end{equation}
%
Therefore the assertion follows also in this case.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:2}
Assume there exists a constant $C_{0}$ such that
$|G^{[n]}(x;\e,c)|\le C_{0}/\al_{n}(\oo)$ for all $n\in\ZZZ_{+}$.
Then there exists $\e_{0}>0$ such that,
for all $c\in B_{r}(c_{0})$ and all $|\e|<\e_{0}$,
the series $c+\overline X(\oo t;\e,c)$ converges.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova Set $D_{0}=\max\{\Gamma,\Phi\}$.
By assumption for all $\theta\in\gotT_{\nn,k}$ one has
%
\begin{eqnarray}
\left| \Val(\theta;\e,c) \right|
& \!\!\! \le \!\!\! &
C_{0}^{k} D_{0}^{k} {\rm e}^{-\xi M(\theta)} 
\Big(\prod_{\ell \in L(\theta)} \al_{n_{\ell}}^{-1}(\oo) \Big) 
\le C_{0}^{k} D_{0}^{k} {\rm e}^{-\xi M(\theta)} \al_{n_{0}}^{-k}(\oo)
\prod_{n=n_{0}+1}^{\io} {\rm e}^{\gotN_{n}(\theta) \log 1/\al_{n}(\oo)}
\nonumber
\\
& \!\!\! \le \!\!\! &
C_{0}^{k} D_{0}^{k} {\rm e}^{-\xi M(\theta)} \al_{n_{0}}^{-k}(\oo)
\exp \left( 4 M(\theta) \sum_{n=n_{0}+1}^{\io} \frac{1}{2^{n}}
\log \frac{1}{\al_{n}(\oo)} \right) ,
\nonumber \end{eqnarray}
%
for arbitrary $n_{0}\in\ZZZ_{+}$. The last sum converges
by Assumption \ref{hyp:1}, so that one can choose $n_{0}$ such that
%
\begin{equation}
\left| \Val(\theta;\e,c) \right| \le 
C_{0}^{k} D_{0}^{k} \al_{n_{0}}^{-k}(\oo) {\rm e}^{-\xi' M(\theta)} ,
\nonumber \end{equation}
%
with $\xi'=\xi/2$. This is enough to prove the lemma.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:3}
For any self-energy cluster $T\in\gotR_{n}$ such that
$\Xi_{n}(x_{T}) \neq 0$, one has
$M(T) \ge 2^{n-1}$ and $\gotN_{p}(T) \le 2^{-(p-2)}M(T)$
for all $p \le n$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova We first prove the bound $M(T) \ge 2^{n-1}$ for $T\in\gotR_{n}$
such that $\Xi_{n}(x_{T}) \neq 0$.
By construction any $T\in\gotR_{n}$ has at least one line $\ell$
with scale $n_{\ell}=n$. If $\ell\notin\calP_{T}$ then $\ell$ is the root
line of a tree $\theta$ such that $\gotN_{n}(\theta)\le
2^{-(n-2)}M(\theta)$ by Lemma \ref{lem:1}, so that
$1 \le \gotN_{n}(\theta) \le 2^{-(n-2)}M(T)$, which yields the bound. 
If all lines with scale $n$ are along $\calP_{T}$ then call
$\ell$ that which is closest to $\ell_{T}^{2}$: by construction
$\ell_{T}^{2}$ and $\ell$ are the entering line and the exiting line,
respectively, of a cluster $T'\subset T$, and
$|\nn_{\ell}-\nn_{\ell_{T}^{2}}| \le M(T')$.
Moreover one has $|\oo\cdot\nn_{\ell}|,|\oo\cdot\nn_{\ell_{T}^{2}}|
\le \al_{n-1}(\oo)/4$, hence $M(T) \ge M(T') \ge
|\nn_{\ell}-\nn_{\ell_{T}^{2}}|\ge 2^{n-1}$.

Now we prove that for $T\in\gotR_{n}$ such that $\Xi_{n}(x_{T})\neq0$
one has $\gotN_{p}(T) \le 2^{-(p-2)}M(T)$ for all $p\le n$.
More generally we prove the bound for the elements
of a wider class of graphs. We say that a subset 
$\widetilde T$ of a tree belongs to the class $\gotS_{n,p}$ if
$\widetilde T$ has one exiting line $\ell_{\widetilde T}^{1}$
and one entering line $\ell_{\widetilde T}^{2}$,
both with scale $\ge p$, and all lines $\ell$ in $\widetilde T'$
have scale $n_{\ell} \le n$. Then we prove the bound
$\gotN_{p}(\widetilde T) \le 2^{-(p-2)}M(\widetilde T)$
for all elements $\widetilde T$ of the class $\gotS_{n,p}$.
The proof is by induction on the number of nodes.
Given a subset $\widetilde T$,
let $\ell_{1},\ldots,\ell_{m}$ the lines on scale $\ge p$ closest
to $\ell_{\widetilde T}^{1}$. If $m=0$ then the bound follows easily.
Also the case in which all lines do not belong to the path
$\calP_{\widetilde T}$ can be easily discussed by relying on
Lemma \ref{lem:1}. If at least one line, say $\ell_{1}$, is along
the path $\calP_{\widetilde T}$, then one has
%
\begin{equation}
\gotN_{p}(\widetilde T) \le 1 +
\gotN_{p}(\widetilde T') + \gotN_{p}(\theta_{2}) + \ldots +
\gotN_{p}(\theta_{m}) , 
\nonumber \end{equation}
%
where $\theta_{i}$, $i=2,\ldots,m$, is the tree with root line
$\ell_{i}$, while $\widetilde T'$ is a subset with the same
properties as $\widetilde T$, i.e. inside the the same class
$\gotS_{n,p}$, but with $N(\widetilde T')<N(\widetilde T)$.
Hence, by the inductive hypothesis,
one has $\gotN_{p}(\widetilde T') \le 2^{-(p-2)}M(\widetilde T')$.
Then the assertion follows once more.
To conclude the proof simply note that if $T\in\gotR_{n}$ then
$T\in\gotS_{n,p}$ for all $p\le n$.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:4}
Assume the propagators $G^{[p]}(x;\e,c)$ are differentiable in $x$
and there exist constants $C_{0}$ and $C_{1}$ such that
$|G^{[p]}(x;\e,c)| \le C_{0}/\al_{p}(\oo)$ and
$|\partial_{x} G^{[p]}(x;\e,c)| \le C_{1}/\al_{p}^{3}(\oo)$ for all $p<n$.
Then there exists $\e_{0}>0$ such that, for all $c\in B_{r}(c_{0})$
and all $|\e|<\e_{0}$, the function $x \mapsto M^{[n]}(x;\e,c)$
is differentiable, and one has
%
\begin{equation}
\left| M^{[n]} (x;\e,c) \right| ,
\left| \partial_{x} M^{[n]} (x;\e,c) \right| \le
D_{1} |\e|^{2} {\rm e}^{-D_{2}2^{n}} ,
\nonumber \end{equation}
%
for some positive constants $D_{1}$ and $D_{2}$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova By proceeding as in the proof of Lemma \ref{lem:2}, one finds
%
\begin{equation}
\left| \Val(T,x;\e,c) \right| \le
C_{0}^{k} D_{0}^{k} {\rm e}^{-\xi M(T)} 
\al_{n_{0}}^{-k}(\oo) \exp
\left( 4 M(T) \sum_{n=n_{0}+1}^{\io} \frac{1}{2^{n}}
\log \frac{1}{\al_{n}(\oo)} \right) ,
\nonumber \end{equation}
%
with $n_{0}$ chosen as in the proof of Lemma \ref{lem:2}.
Then one can use Lemma \ref{lem:3} to bound $M(T)$,
and the observation that any self-energy cluster $T$
has at least two nodes to obtain the factor $\e^{2}$.
This proves the bound on $M^{[n]} (x;\e,c)$.

To obtain the bound on $\partial_{x}M^{[n]} (x;\e,c)$ simply note that
%
\begin{equation}
\partial_{x} M^{[n]}(x;\e,c) = \sum_{T\in\gotR_{n}}
\Big( \prod_{v \in E(T) \cup V(T)} F_{v} \Big)
\sum_{\ell\in \calP_{T}} \partial_{x} G_{\ell}
\Big(\prod_{\ell' \in L(\theta) \setminus\{\ell\}} G_{\ell'} \Big) ,
\nonumber \end{equation}
%
where $\partial_{x} G_{\ell}$ can be bounded as $|\partial_{x}
G_{\ell}| \le C_{1}/\al_{n_{\ell}}^{3}(\oo)$ by hypothesis.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:5}
%Under the same assumptions as in Lemma \ref{lem:3},
Assume there exists a constant $C_{0}$ such that
$|G^{[p]}(x;\e,c)|\le C_{0}/\al_{p}(\oo)$ for all $p<n$.
Then one has $(\MM^{[p]}(x;\e,c))^{*} = \MM^{[p]}(-x;\e,c)$
for all $p \le n$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova The proof is by induction on $p$. First of all note that
if $(\MM^{[p]}(x;\e,c))^{*} = \MM^{[p]}(-x;\e,c)$ then
$(G^{[p]}(x;\e,c))^{*} = G^{[p]}(-x;\e,c)$ by (\ref{eq:3.7a}).
Moreover one has $F_{v}^{*}=F_{v}$ for all internal nodes $v\in V(T)$ and
$F_{v}^{*}=f_{\nn_{v}}^{*}=f_{-\nn_{v}}$ for all end nodes $v\in E(T)$.

Let $T$ a self-energy cluster contributing to $\MM^{[p]}(x;\e,c)$
-- see (\ref{eq:3.7b}) -- for $p\le n$; then $T\in \gotR_{q}$ for
some $q\le p$. Together with $T\in\gotR_{q}$ consider also the
self-energy cluster $T'\in\gotR_{q}$ obtained from $T$ by changing
the signs of the mode labels of all the end nodes $v\in E(T)$.
Note that there is a one-to-one correspondence between
the self-energy clusters $T$ and $T'$.
The node factors corresponding to the end nodes $v \in E(T')$
become $f_{-\nn_{v}}$, and, if we revert the momentum of the
entering line $\ell_{T'}^{2}$, the momenta of all the lines
$\ell\in L(T')$ also change sign, that is $\nn_{\ell}$ is replaced
with $-\nn_{\ell}$ for all $\ell\in L(T')$.

The definition (\ref{eq:3.8}) and the inductive hypothesis
yield $(\Val(T,x;\e,c))^{*}=\Val(T',-x;\e,c)$ for all $q \le p$
and all $T\in\gotR_{q}$. Then (\ref{eq:3.7b}) implies the assertion.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:6}
For all $n\in\ZZZ_{+}$ the function $x \mapsto \MM^{[n]}(x;\e,c))$
is differentiable and one has
$|\ii x (1 + \ii \e x) - \MM^{[n]}(x;\e,c)|\ge |x|/2$
for all $c\in B_{r}(c_{0})$ and all $\e$ small enough.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova The proof is by induction on $n$. Assume that
the functions $x \mapsto \MM^{[p]}(x;\e,c)$ are differentiable and
one has $|\ii x (1 + \ii \e x) - \MM^{[p]}(x;\e,c)|\ge |x|/2$
for all $p<n$. One can easily verify that then also the propagators
$G^{[p]}(x;\e,c)$ are differentiable and satisfy the bounds
$|\partial_{x} G^{[p]}(x;\e,c)| \le C_{1}/\al_{p}^{3}(\oo)$ 
for all $p \le n$ and for some positive constant $C_{1}$. Indeed one has
%
\begin{equation}
\partial_{x} G^{[p]}(x;\e,c) =
\frac{\partial_{x}\Psi_{p}(x)}{\ii x(1+\ii \e x)-\MM^{[p-1]}(x;\e,c)} -
\frac{\Psi_{p}(x) \left( \ii - 2\e x -\partial_{x}\MM^{[p-1]}
(x;\e,c) \right)}{(\ii x(1+\ii \e x)-\MM^{[p-1]}(x;\e,c))^{2}} ,
\nonumber \end{equation}
%
where
%
\begin{eqnarray}
\partial_{x}\Psi_{p}(x)
& \!\!\! = \!\!\! & \sum_{j=0}^{p-1}
\chi_{0}(|x|) \ldots \partial_{x} \chi_{j}(|x|) \ldots \psi_{p}(|x|) +
\chi_{0}(|x|) \ldots \chi_{p-1}(|x|) \partial_{x} \psi_{n}(|x|)
\nonumber \\
& \!\!\! \le \!\!\! &
C \sum_{j=0}^{p} \al_{j}^{-1}(\oo) \le C p \al_{p}^{-1}(\oo)
\nonumber \end{eqnarray}
%
for some constant $C$, and
%
\begin{eqnarray}
\partial_{x}\MM^{[p-1]}(x;\e,c)
& \!\!\! = \!\!\! & \sum_{j=0}^{p-1} \left( \sum_{i=0}^{j} 
\chi_{0}(|x|) \ldots \partial_{x} \chi_{i}(|x|) \ldots \chi_{j}(|x|)
M^{[j]}(x;\e,c) + \Xi_{j}(x) \partial_{x} M^{[j]}(x;\e,c) \right)
\nonumber \\
& \!\!\! \le \!\!\! &
C |\e|^{2}  \sum_{j=0}^{p-1} {\rm e}^{-D_{2}2^{j}}
\left( j \al_{j}^{-1}(\oo) + 1 \right) \le
C' |\e|^{2} p^{2} \al_{p}^{-1}(\oo) ,
\nonumber \end{eqnarray}
%
for some constants $C,C'$.

Then we can apply Lemma \ref{lem:4} to conclude that $\MM^{[n]}(x;\e,c)$
is differentiable and its derivative with respect to $x$
is accordingly bounded. Therefore
%
\begin{equation}
\ii x (1 + \ii \e x) - \MM^{[n]}(x;\e,c) =
\ii x (1 + \ii \e x) - \MM^{[n]}(0;\e,c) -
\left( \MM^{[n]}(x;\e,c) - \MM^{[n]}(0;\e,c) \right) ,
\nonumber \end{equation}
%
where $\MM^{[n]}(0;\e,c)$ is real by Lemma \ref{lem:6}, and
%
\begin{equation}
\left| \MM^{[n]}(x;\e,c)) - \MM^{[n]}(0;\e,c) \right|
\le C |\e|^{2} |x| ,
\nonumber \end{equation}
%
for some constant $C$, by Lemma \ref{lem:4}.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:7}
Then there exists $\e_{0}>0$ such that,
for all $c\in B_{r}(c_{0})$ and all $|\e|<\e_{0}$, the function
$c+\overline X(\oo t;\e,c)$ solves (\ref{eq:2.3}).
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova We have to prove that the coefficients $\overline X_{\nn}$,
defined abstractly through (\ref{eq:3.10}), solve the
first equation in (\ref{eq:2.2}), i.e.
%
\begin{equation}
\left( \ii\oo\cdot\nn \right)
\left( 1 + \ii\e\oo\cdot\nn \right) \overline X_{\nn} +
\e [g \circ (c + \overline X)]_{\nn} = \e f_{\nn} ,
\qquad \nn \neq \vzero .
\nonumber \end{equation}
%
Set $D_{n}(x;\e,c)= \ii x \left( 1+\ii\e x \right)-\MM^{[n]}(x;\e,c)$,
so that $G^{[n]}(x;\e,c)= \Psi_{n}(x)/D_{n}(x;\e,c)$, and
$G(x)=1/(\ii x)(1+\ii\e x)$. Write also
%
\begin{equation}
\overline X_{\nn} = \sum_{n=0}^{\io} \overline X_{\nn,n} ,
\qquad \overline X_{\nn,n} =
\sum_{k=1}^{\io} \e^{k} \sum_{\theta\in\gotT_{k,\nn,n}}
\Val(\theta;\e,c) ,
\nonumber \end{equation}
%
where $\gotT_{k,\nn,n}$ is the subset of $\gotT_{k,\nn}$
of the renormalised trees with root line with scale $n$.

If we define
%
\begin{equation}
\Omega(\nn,\e,c) =
G(\oo\cdot\nn) \left[ \e f - \e g(c +
\overline X(\cdot;\e,c) \right]_{\nn} ,
\label{eq:3.13} \end{equation}
%
then we have to prove that $\Omega(\nn,\e,c) = \overline X_{\nn}$.

By setting
%
\begin{equation}
\Psi_{j,n}(x) = \chi_{j}(|x|)\ldots \chi_{n-1}(|x|) \psi_{n}(|x|) ,
\quad n>j , \qquad \qquad \Psi_{n,n}(x) = \psi_{n}(|x|) , 
\label{eq:3.14} \end{equation}
%
note that
%
\begin{equation}
\Psi_{0,n}(x) = \Psi_{n}(x), \qquad \qquad
\sum_{n=j}^{\io} \Psi_{j,n}(x) = 1 \quad \forall j \ge 0 .
\label{eq:3.15} \end{equation}
%
Then, by using the last identity in (\ref{eq:3.15}) with $j=0$,
we can rewrite (\ref{eq:3.13}) as
%
\begin{equation}
\Omega(\nn,\e,c) =
G(\oo\cdot\nn) \sum_{n=0}^{\io}
D_{n}(\oo\cdot\nn;\e,c) \, G^{[n]}(\oo\cdot\nn;\e,c)
\left[ \e f - \e g(c + \overline X(\cdot;\e,c)) \right]_{\nn} ,
\label{eq:3.16} \end{equation}
%
where we can expand
%
\begin{eqnarray}
& & G^{[n]}(\oo\cdot\nn;\e,c)
\left[ \e f_{\nn} - \e g(c + \overline X(\cdot;\e,c)) \right]_{\nn} =
\sum_{k=1}^{\io} \e^{k} \sum_{\theta\in\gotT_{k,\nn,n}} \Val(\theta;\e,c)
\nonumber \\
& & \qquad \qquad
+ \; G^{[n]}(\oo\cdot\nn;\e,c)
\sum_{p=n}^{\io} \sum_{j=0}^{n-1} M^{[j]}(\oo\cdot\nn;\e,c)
\sum_{k=1}^{\io} \e^{k} \sum_{\theta\in\gotT_{k,\nn,p}} \Val(\theta;\e,c)
\nonumber \\
& & \qquad \qquad
+ \; G^{[n]}(\oo\cdot\nn;\e,c)
\sum_{p=0}^{n-1} \sum_{j=0}^{p-1} M^{[j]}(\oo\cdot\nn;\e,c)
\sum_{k=1}^{\io} \e^{k} \sum_{\theta\in\gotT_{k,\nn,p}} \Val(\theta;\e,c) ,
\nonumber \end{eqnarray}
%
where the sum in the second line is present only if $n\ge 1$
and the sum in the third line is present only if $n\ge 2$.
Therefore we obtain
%
\begin{eqnarray}
& & \Omega(\nn,\e,c) =
G(\oo\cdot\nn) \sum_{n=0}^{\io} D_{n}(\oo\cdot\nn;\e,c)
\overline X_{\nn,n}
\nonumber \\
& & \qquad \qquad
+ \; G(\oo\cdot\nn) \sum_{n=1}^{\io} \Psi_{n}(x)
\sum_{p=n}^{\io} \sum_{j=0}^{n-1} M^{[j]}(\oo\cdot\nn;\e,c)
\overline X_{\nn,p}
\nonumber \\
& & \qquad \qquad
+ \; G(\oo\cdot\nn) \sum_{n=2}^{\io} \Psi_{n}(x)
\sum_{p=0}^{n-1} \sum_{j=0}^{p-1} M^{[j]}(\oo\cdot\nn;\e,c)
\overline X_{\nn,p} .
\nonumber \end{eqnarray}
%
The second and third lines, summed together, give
%
\begin{equation}
G(\oo\cdot\nn) \sum_{n=1}^{\io} \overline X_{\nn,n}
\sum_{j=0}^{n-1} M^{[j]}(\oo\cdot\nn;\e,c)
\sum_{p=j+1}^{\io} \Psi_{p}(x) , \qquad \hbox{where} \qquad
\sum_{p=j+1}^{\io} \Psi_{p}(x) = \Xi_{j}(x) ,
\nonumber \end{equation}
%
where we have written $\Psi_{p}=\Xi_{j}\Psi_{j+1,p}$
and used (\ref{eq:3.14}) to obtain the last equality, so that
(\ref{eq:3.16}) gives
%
\begin{equation}
\Omega(\nn,\e,c) =
G(\oo\cdot\nn) \sum_{n=0}^{\io}
\left( D_{n}(\oo\cdot\nn;\e,c) + \MM^{[n]}(\oo\cdot\nn;\e,c) 
\right) \overline X_{\nn,n} = \sum_{n=0}^{\io}
\overline X_{\nn,n} = \overline X_{\nn} ,
\nonumber \end{equation}
%
which proves the assertion.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:8}
The function $\overline X(\pps;\e,c)$ is $C^{\io}$
in $\e$ and $c$ for $\e$ and $c-c_{0}$ small enough.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova The previous results imply that $\overline X(\cdot;\e,\cdot)$
is a well defined function of $\e$ for $\e$ small enough. 
By looking at the tree expansion (\ref{eq:3.9}) for the coefficients
$\overline X^{[k]}_{\nn}$ of  $\overline X(\pps;\e,c)$,
one sees that the function depends on $\e$ through the
factors $\e^{k}$ in (\ref{eq:3.10}) and through the propagators
$G_{\ell}$. The first dependence is trivial, and poses no obstacle
in differentiating. Also the dependence through the propagators
can be easily handled thanks to Lemma \ref{lem:6}, which allows
to bound from below the denominators. In particular
for all $m \ge 0$ one finds
%
\begin{equation}
\left| \partial_{\e}^{m} G^{[p]}(x;\e,c) \right| \le K_{m} /\al_{p}^{m}(\om)
\nonumber \end{equation}
%
for suitable constants $K_{m}$. Smoothness in $c$ can be discussed
in a similar way, by using analyticity of the force $g$ and again
Lemma \ref{lem:6}. \EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{The implicit function equation}
\label{sec:4}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We are left with the implicit function equation (\ref{eq:2.4}), which
can be trivially solved under Assumption \ref{hyp:2}. If we define
%
\begin{equation}
\Gamma(\e,c) = [g(c+X(\cdot;\e,c)]_{\vzero} - f_{\vzero} ,
\label{eq:4.1} \end{equation}
%
then the following result holds.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:9}
There exists a neighbourhood $U \times V$ of $(\e,c)=(0,c_{0})$
such that for all $\e\in U$ there is at least one value $c=c(\e) \in V$,
depending continuously on $\e$, for which one has $\Gamma(\e,c(\e))=0$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova Since $\Gamma(0,c)=g(c)-f_{\vzero}$,
Assumption \ref{hyp:2} implies that
%
\begin{equation}
\frac{\der^{k}}{\der c^{k}} \Gamma(0,c_{0})=0 \quad
\hbox{for} \quad k=0,1,\ldots,\gotn -1 \quad \hbox{and} \quad
\Gamma_{0} = \frac{\der^{\gotn}}{\der c^{\gotn}} \Gamma(0,c_{0}) \neq 0 .
\label{eq:4.2} \end{equation}
%
Set $\s_{0}={\rm sign}(\Gamma_{0})$ so that $\s_{0}\Gamma_{0}>0$.
By continuity there are neighbourhoods $U$ and $V=[V_{-},V_{+}]$ of
$\e=0$ and $c=c_{0}$, respectively, such that for all $\e\in U$ one has
$\s_{0}\Gamma(\e,c)>0$ for $c =V_{+}$ and $\s_{0}\Gamma(\e,c)<0$
for $c =V_{-}$. Therefore, there exists a continuous curve $c=c(\e)$
such that $\Gamma(\e,c(\e))=0$.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

By collecting together the results of the previous sections and
Lemma \ref{lem:9}, Theorem \ref{thm:1} follows.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{Zeroes of even order}
\label{sec:5}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section we prove the following result, which, together with
Theorem \ref{thm:1}, implies Theorem \ref{thm:2}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{lemma} \label{lem:10}
Under Assumption \ref{hyp:1} for the ordinary differential equation
(\ref{eq:1.1}), assume also that $c_{0}$ is a zero of even order
of (\ref{eq:1.1}). Then there is no quasi-periodic solution
reducing to $c_{0}$ when $\e$ tends to $0$.
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\prova The analysis of Section \ref{sec:3} shows that a solution
of the range equation (\ref{eq:2.3}) can be proved to exist
under the only Assumption \ref{hyp:1}. Moreover such a solution
is $C^{\io}$ in both $\e$ and $c$ (cf. Lemma \ref{lem:8}).
Then, we study the bifurcation equation (\ref{eq:2.4}) in the case
$c_{0}$ is a zero of even order of (\ref{eq:1.4}).

If we write $c=c_{0}+\zeta$ and expand the function $g \circ (c+X)$
around $c=c_{0}$, then (\ref{eq:2.4}) gives
%
\begin{equation}
[g(c+X(\cdot;\e,c)]_{\vzero} - f_{\vzero} =
g_{0} \laa \left( \zeta + X \right)^{\gotn} \raa + 
\laa O\left( \zeta + X \right)^{\gotn+1} \raa = 0 ,
\label{eq:5.1} \end{equation} 
%
where $\gotn! g_{0}=\der^{\gotn}g/\der x^{\gotn}(c_{0})\neq 0$
and $\laa\cdot\raa$ denotes as usual the Fourier component
with label $\nn=\vzero$. If $\e=O(\zeta)$ then we have
$|g_{0}| \laa \left( \zeta + X \right)^{\gotn} \raa \ge C_{1}\e^{\gotn}$
for some positive constant $C_{1}$, because $\gotn$ is even,
and $O\left( \zeta + X \right)^{\gotn+1} = O(\e^{\gotn+1})$,
so that (\ref{eq:5.1}) cannot be satisfied for $\e$ small enough.

If $\e=o(\zeta)$, then
%
\begin{equation}
\laa \left( \zeta + X \right)^{\gotn} \raa =
\sum_{k=0}^{\gotn} \left( \begin{matrix} \gotn \\ k \end{matrix} \right)
\zeta^{k} \laa X^{\gotn - k} \raa = \zeta^{\gotn} + o(\zeta^{\gotn})
\label{eq:5.2} \end{equation} 
%
for $\e$ small enough. On the other hand
$O\left( \zeta + X \right)^{\gotn+1} = O(\zeta^{\gotn+1})$,
and hence once more there is no solution to (\ref{eq:5.1})
because of (\ref{eq:5.2}).
The case $\zeta=o(\e)$ can be discussed in a similar way.\EP
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\zerarcounters
\section{Conclusions and open problems}
\label{sec:6}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The analysis of the previous sections shows that under Assumptions
\ref{hyp:1} and \ref{hyp:2} the system described by the ordinary
differential equation (\ref{eq:1.1}) admits a response solution.
Under some mild conditions on $g$ one can prove that
such a solution describes a (local) attractor \cite{BDG}.
It would be interesting to investigate whether the same
result can be obtained by only making Assumption \ref{hyp:2} on $g$
and requiring $\der^{\gotn}g/\der x^{\gotn}(c_{0})>0$.
Even more interesting would be to understand whether the same scenario
persists after removing Assumption \ref{hyp:1} on $\oo$.
The analysis of \cite{BDG} shows that, if there is a quasi-periodic
solution of the form considered in Theorem \ref{thm:1} exists,
then it is an attractor (under some conditions on $g$), but
if $\oo$ does not satisfy any Diophantine condition,
such as the Bryuno condition, then the small divisor problem
can not be handled, and it is very unlikely that the dynamics
can be conjugated to the unperturbed one.

The analysis in Section \ref{sec:5} shows
that, if $c_{0}$ is a zero of even order $\gotn$ for the
equation (\ref{eq:1.4}), then no quasi-periodic
solution of the form considered in Theorem \ref{thm:1} exists.
A natural question in that case is, how the dynamics evolves in time,
and what kind of attractors arise.

Furthermore, Theorem \ref{thm:1} states that for all
$\e$ small enough there is a value $c(\e)$ for the average
of $x(t)$, such that the solution exists, but provides nothing
more than continuity about the dependence of $c(\e)$ on $\e$.
Thus, another question which should deserve further investigation
is, if under some further assumption one can prove some
stronger regularity property for the function $c(\e)$ --
note that analyticity fails to hold even in the case of
periodic forcings \cite{GBD1}. In this direction,
the results of \cite{CG} could provide a possible guideline
(even if in this case the implicit function equation
to be studied is no longer analytic), not only to prove
smoothness but also to provide an algorithm to explicitly
construct the function $c(\e)$. Of course,
under the Assumption \ref{hyp:2} on $\oo$,
independently of the conditions on $g$, we have no hope
to prove Borel summability \cite{S} in $\e$ at the origin.
Indeed, this should require a much stronger Diophantine condition
on $\oo$ \cite{CGGG,GBD2}. 

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% References
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\end{document}
